Author's response to reviews

Title: Intestinal permeability - a new target for disease prevention and therapy?

Authors:

Stephan C Bischoff (bischoff.stephan@uni-hohenheim.de)
Giovanni Barbara (giovanni.barbara@unibo.it)
Wim Buurman (w.buurman@maastrichtuniversity.nl)
Theo Ockhuizen (ockhuizen@nutricom.nl)
Jörg-Dieter Schulzke (joerg.schulzke@charite.de)
Matteo Serino (matteo.serino@inserm.fr)
Herbert Tilg (herbert.tilg@i-med.ac.at)
Alastair Watson (Alastair.Watson@uea.ac.uk)
Jerry Wells (jerry.wells@wur.nl)

Version: 2
Date: 14 January 2014

Author's response to reviews:

MS: 7967014621171251
Debate

Intestinal permeability - a new target for disease prevention and therapy?
Stephan C Bischoff, Giovanni Barbara, Wim Buurman, Theo Ockhuizen, Jörg-Dieter Schulzke, Matteo Serino, Herbert Tilg, Alastair Watson and Jerry Wells

BMC Gastroenterology (Section: Intestinal disorders)

Dear Editor,

thank you for your email dated Dec 28th, 2013. Because of vacation and some temporary problems in accessing your website, we could not answer earlier.

You asked us to to make some changes to the formatting of our manuscript.
(1) We included now email addresses of each author in the title page.
(2) We included now an Acknowledgment section.
(3) Ethics statement. This is not applicable in our case. Since our manuscript is a debate, no original human material or data was used and no approval of an ethics committee exists.

We uploaded the revised version.

Thank you very much!

Kind regards,

Stephan
Stephan C. Bischoff, M.D.
Professor of Gastroenterology and Nutritional Medicine